Author's response to reviews

Title: The effect of concentrated bone marrow aspirate in operative treatment of fifth metatarsal stress fractures; a double-blind randomized controlled trial

Authors:

Hanneke Weel (h.weel@amc.nl)
Wouter H Mallee (w.h.mallee@amc.nl)
C Niek van Dijk (c.n.vandijk@amc.nl)
Leendert Blankevoort (l.blankevoort@amc.nl)
Simon Goedegebuure (simongoedegebuure@desportartsengroep.nl)
J Carel Goslings (j.c.goslings@amc.nl)
John G Kennedy (kennedyj@hss.edu)
Gino MMJ Kerkhoffs (g.m.kerkhoffs@amc.nl)

Version: 6 Date: 15 July 2015

Author's response to reviews:

Dear editors,

These minor formatting changes we performed for the manuscript "The effect of concentrated bone marrow aspirate in operative treatment of fifth metatarsal stress fractures; a double-blind randomized controlled trial":
- the version date in the beginning of the manuscript is removed
- the track-change-version is removed
- abbreviations section is placed before the competing interests section.

We hope the manuscript will be now considered for publication in your journal. Looking forward for your decision.

Yours sincerely,

Hanneke Weel and other authors.